Author’s response to reviews

Title: Influence of iRoot SP and mineral trioxide aggregate on the activation and polarization of macrophages induced by lipopolysaccharide

Authors:

Zhenglin Yuan (zldentist@hotmail.com)
Xiaodan Zhu (308859980@qq.com)
YuHong Li (1004809372@whu.edu.cn)
Ping Yan (yanpingjh@hotmail.com)
Han Jiang (56496675@qq.com)

Version: 4 Date: 04 Mar 2018

Author’s response to reviews:

1. We have attached the iThenticate report the areas of highest overlap. Please alter these areas, especially in the abstract. The areas of concern are those highlighted by a red number 1.

I have revised all the red color areas to avoid the overlap.

2. We would also ask that you make reference to your previous work in the Background of this manuscript and clarify how this current manuscript differs from your previous piece of published research.

To clarify the difference between the current manuscript and our previous manuscript, on page 7, from line 257 to line 271, some new contents have been added to introduce between the current study and our previous study.